Author’s response to reviews

Title: The experience of Community First Responders in co-producing rural health care: in the liminal gap between citizen and professional

Authors:

Anne Anne Roberts Mrs (a.roberts@abdn.ac.uk)
Amy Nimegeer Dr (amy.nimegeer@stir.ac.uk)
Jane Farmer Professor (j.farmer@latrobe.edu.au)
David J Heaney Mr (d.heaney@abdn.ac.uk)

Version: 2 Date: 12 November 2013

Author’s response to reviews:

Dear BMC Health Services Research,

We have made the requested amendments to our article and I attach our revised manuscript for your review.

1. The title page now includes a list of full authors, highlighting the corresponding author
2. We have changed the heading “findings” to “results” as advised.
3. Vertical lines have been removed from the table in our manuscript
4. We have declared we have no conflict of interest.
5. We have made acknowledgements in this paper
6. We have included abbreviations after the conclusion section.
7. We have detailed author contributions.

I hope you enjoy reading our manuscript.

Kind Regards
Mrs Anne Roberts
Corresponding Author